Author's response to reviews

Title: Predicting prognosis in hepatocellular carcinoma after curative surgery with common clinicopathologic parameters

Authors:

Ke Hao (ke_hao@merck.com)
John M Luk (jmluk@nus.edu.sg)
Nikki P Y Lee (nikkilee@hkucc.hku.hk)
Mao Mao (mao_mao@merck.com)
Chunsheng Zhang (Chunsheng_Zhang@merck.com)
Mark D Ferguson (Mark_Ferguson@merck.com)
John Lamb (John_Lamb3@merck.com)
Hongyue Dai (hongyue_dai@merck.com)
Irene O Ng (iolng@hkusua.hku.hk)
Pak C Sham (pcsham@hkucc.hku.hk)
Ronnie T P Poon (poontp@hkucc.hku.hk)

Version: 7 Date: 28 October 2009

Author's response to reviews:

Dear Production team,

There are few minor revisions as follows:

Title page, affiliation 5: add National University of Singapore
Page 5, bottom line 6: time-depend to time-dependent
Page 6, bottom line 5: reserved changed to reserve
Page 11, bottom line 2: parameters, changed to parameters;
Page 12, bottom line 7: manuscript, changed to manuscript.
Page 15, Log-rank curves changed to Kaplan-Meier survival curves
Page 17, Table 1 title: clinic changed to clinicopathologic

Thank you for your prompt attention.

John Luk